Author's response to reviews

Title: A novel observation of pubic osteomyelitis due to Streptococcus viridans after dental extraction: a case report

Authors:

Naseem Naqvi (naseemnaqvi@doctors.org.uk)
Rizwana Naqvi (drriz00@yahoo.com)
Christopher Wong (Christopher.Wong@aintree.nhs.uk)
Sushmita Pearce (Sushmita.Pearce@wwl.nhs.uk)

Version: 9  Date: 18 May 2008

Author's response to reviews:

Changes made as requested, in the authors details:

Naseem Naqvi*1, Rizwana Naqvi2, Christopher Wong3 and Sushmita Pearce4

1 Department of Acute Medicine, Dumfries & Galloway Royal Infirmary, Scotland, UK.
2 Department of Medicine, Macclesfield District General Hospital, East Cheshire NHS Trust, UK.
3 Department of Renal Medicine, Aintree University Hospitals NHS Trust, Liverpool, UK.
4 Department of Medicine, Royal Albert Edward Infirmary, Wigan, UK.